Author’s response to reviews

Title: Dissecting the signaling pathways associated with the oncogenic activity of MLK3 P252H mutation

Authors:

Sergia Velho (svelho@ipatimup.pt)
Ana Pinto (apinto@ineb.up.pt)
Danilo Licastro (danilo.licastro@cbm.fvg.it)
Maria J Oliveira (mariojo@ineb.up.pt)
Filipa Sousa (filipa.sousa@gmail.com)
Elia Stupka (stupka.elia@hsr.it)
Raquel Seruca (rseruca@ipatimup.pt)

Version: 4 Date: 14 February 2014

Author’s response to reviews: see over
Dear Dr. Cherry Battad
Journal Editorial Office
BioMedCentral

In order to reply to the Editorial request, we have now performed the microarray data deposition at the GEO repository. The accession number is: GSE54611.

Looking forward to hearing from you,

Professor Raquel Seruca
Coordinator - Cancer Genetics Group
Vice-President of IPATIMUP